Author's response to reviews

Title: Prevalence of smoking during pregnancy and associated risk factors among Canadian women: a national survey

Authors:

Ban Al-Sahab (bsahab@yorku.ca)
Masarat Saqib (masarat@yorku.ca)
Gabriel Hauser (ghauser@yorku.ca)
Hala Tamim (htamim@yorku.ca)

Version: 7 Date: 19 May 2010

Author's response to reviews:

The following minor changes have been made:

Page 1: Email addresses: Changed "BA-S" to "BAS"
Page 2: 4th line: Changed "it is" to "its"
Page 4: 6th line: added a comma after the phrase: "To our knowledge"
Page 6: 9th line: changed "cigarette" to "cigarettes"
Page 8: 5th line from the bottom: Added a comma before the word "respectively"
Page 10: 5th line from the bottom: Added the word "in" before the word "Ontario"
Page 11: 3rd line: added a comma after the phrase "In another study"
Page 11: 7th line from the bottom: deleted the word "in" before the word "Manitoba"